Author's response to reviews

Title: Podocyte specific knock out of selenoproteins does not enhance nephropathy in streptozotocin diabetic C57BL/6 mice

Authors:

Marsha N Blauwkamp (mousman@umich.edu)
Jingcheng Yu (jingyu@umich.edu)
MaryLee A Schin (mschin@umich.edu)
Kathleen A Burke (kathleen_burke@brown.edu)
Marla J Berry (mberry@hawaii.edu)
Bradley A Carlson (carlsonb@mail.nih.gov)
Frank C Brosius III (fbrosius@umich.edu)
Ronald J Koenig (rkoenig@umich.edu)

Version: 3  Date: 17 July 2008

Author's response to reviews:

Abstract was updated in the Manuscript details tab to match that in the actual manuscript.
References were updated to include all authors (less than 30), thus eliminating the use of et al.
Acknowledgments were updated.
Figure 1 was cropped to eliminate white space surrounding the illustration.